Author's response to reviews

Title: Cross-sectional associations between prevalent vertebral fracture and pulmonary function. The sixth Tromso Study

Authors:

Bente Morseth (bente.morseth@uit.no)
Hasse Melbye (hasse.melbye@uit.no)
Svanhild Waterloo (svanhild.h.waterloo@uit.no)
Marte R Thomassen (marte.r.thomassen@uit.no)
Marijke J Risberg (Marijke.E.Jongsma.Risberg@unn.no)
Nina Emaus (nina.emaus@uit.no)

Version: 4 Date: 21 October 2013

Author's response to reviews:

In this revision, we have clarified the copyright details for reprinting Figure 2. The copyright holder for the figure is the American Journal of Roentgenology. We have stated in the manuscript (Figure 2 legend) that permission was obtained from the American Journal of Roentgenology.